\begin{table}%t5
    %\centering
\par
\caption{References to ISOCAM sources in existing archives. The first column lists the source number from
Tables~\ref{lw3} and \ref{lw3_add}. The key to the references in Cols.~2 to~8 is Dressler \& Gunn
(\cite{1992ApJS...78....1D}, DG), Czoske et~al. (\cite{2001AetA...372..391C}, CKS), Schneider et~al.
(\cite{1986AJ.....92..523S}, SDG), Smail et~al. (\cite{1997ApJS..110..213S}, SDC), McLean \& Teplitz
(\cite{1996AJ....112.2500M}, MT), Butcher \& Oemler (\cite{1978ApJ...219...18B}, BO), Soucail et~al.
(\cite{2000AetA...355..433S}, S) and Pickles \& van der Kruit (\cite{1991AetAS...91....1P}, P). The table does not
include the four stars or the quasar PC 0023+1653.\label{other}}
   \begin{tabular}{cccccccl}
\hline \hline
id.    & DG & CKS & SDG & SDC & MT  & BO  &Others\\
\hline \noalign{\smallskip}
05    &198 &262  &223  &     &     &123  &      \\
06    &278 &     &     &     &64   &     &      \\
08    &264 &267  &113  &     &22   &34   &S01   \\
09    &257 &282  &     &     &6    &     &      \\
10    &237 &289  &     &     &     &     &      \\
12    &    &381  &     &     &     &     &      \\
17    &12  &     &     &     &     &     &      \\
18    &47  & 444 &     &797  &4    &     &      \\
21    &119 &453  &     &     &     &     &      \\
22    &43  &459  &146  &834  &     &     &      \\
23    &23  &     &     &883  &     &     &      \\
24    &    &471  &     &     &     &     &      \\
27    &3   &474  &     &928  &     &     &      \\
1s    &195 &280  &     &     &50   &     &      \\
3s    &246 &304  &     &     &55   &     &      \\
4s    &101 &412  &186  &573  &11   &87   &P61   \\
 \hline
      \end{tabular}
\end{table}